Author's response to reviews

Title: Extraperitoneal urine leak after renal transplantation: the role of radionuclide imaging and the value of accompanying SPECT/CT - a case report

Authors:

Hongju Son (hongju.son@mssm.edu)
Sherif Heiba (sherif.heiba@mssm.edu)
Lale Kostakoglu (lale.kostakoglu@mssm.edu)
Josef Machac (josef.machac@mssm.edu)

Version: 6 Date: 21 June 2010

Author's response to reviews: see over
June 18, 2010

Dear Tim Shipley, PhD:

Below are our responses regarding your comments.

Major revisions (we require the author to make these changes)

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Case Presentation, Conclusions → It was reformatted.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Reference 6 is currently cited before reference 5. → It was corrected.

We would greatly appreciate your comments.

Sincerely yours,

Hongju Son, MD
Mount Sinai Medical Center
Nuclear Medicine Resident